\begin{table}%t4
%\centering
\par
\caption{\label{tab4}Input data for the curves-of-growth of M~giants based on PP~model (results based on SS models in parenthesis).}
\small \begin{tabular}{l c l c c c l l}
\hline \hline
\noalign{\smallskip}
Object & Molecule & PP model & $\xi_{\rm micro}^{\rm PP}$ 
($\xi_{\rm micro}^{\rm SS}$) & log~$A_{\rm C}^{\rm PP}$ 
(log~$A_{\rm C}^{\rm SS}$) & log~$A_{\rm O}^{\rm PP}$ 
(log~$A_{\rm O}^{\rm SS}$) &  CG analysis & LL analysis  \\[4mm]
       &   &($T_{\rm eff}$/log~$g$)  &  (km~s$^{-1}$) &
 &   & (this Paper) & (Tsuji \cite{Tsuji08})    \\
\hline
$\beta$ Peg & CO & 3600/0.5 & 1.88 $\pm$ 0.07 (2.04) & 
8.17 $\pm$ 0.05 (8.27)& -- &  Fig.~\ref{Fig6}a & Fig.~\ref{Fig6}a,b \\
$\rho$ Per  & CO  &  3500/0.5 & 3.52 $\pm$ 0.73 (3.59) & 8.17 $\pm$ 0.08
(8.27) & --  & Fig.~\ref{Fig7}a & Fig.~\ref{Fig8}a,b \\
$\alpha$ Her  & CO & 3300/0.5 & 2.67 $\pm$ 0.71 (2.82) & 
8.46 $\pm$ 0.16 (8.40) & -- & Fig.~\ref{Fig6}b & Fig.~\ref{Fig4}a,b \\
SW Vir  & CO & 2900/0.0 & 4.10 $\pm$ 3.00 (4.21) & 
8.22 $\pm$ 0.22 (8.26) & -- &  Fig.~\ref{Fig7}b & Fig.~\ref{Fig9}c \\ 
$\alpha$ Her  & OH & 3300/0.5 & 2.57 $\pm$ 0.09 (2.92) & -- & 
8.82 $\pm$ 0.05 (8.85) &  Fig.~\ref{Fig8}a & Fig.~\ref{Fig10}a,b  \\
SW Vir   &   OH  &  2900/0.0  & 1.69 $\pm$ 0.15 (1.83)  & -- & 
8.42 $\pm$ 0.12 (8.54) & Fig.~\ref{Fig8}b & Fig.~\ref{Fig11}c\\ 
\hline
\end{tabular}
\end{table}